Author's response to reviews

Title: Mopane worm allergy: A Case Report

Authors:

okechukwu A okezie (okeyokezie@yahoo.com)
koloi k kgomotso (kkoloi@bhp.org.bw)
Mavis M letswiti (mletswiti@bhp.org.bw)

Version: 4 Date: 27 November 2008

Author's response to reviews:

I have added that the patient had no previous history of reaction to mompane worm and no allergic illness in the past. The patients symptoms on presenting to the clinic and more information on the drugs used in treating patients have also been added.